Author's response to reviews

Title: A novel mitochondrial DNA mutation associated with hypertension in C4329A mutation between tRNA(Ile) and tRNA(Grn) genes

Authors:

yuqi liu (ametuof4290@gmail.com)
jinliao gao (gaojl@163.com)
ynfeng lan (lanvf@163.com)
zongbin li (zongbl@163.com)
chao zhu (zhuchao@163.com)
yang li (yangli@163.com)
jie yang (yangjie@gmail.com)
xinmin guan (minxg@163.com)

Version: 4 Date: 2 December 2013

Author's response to reviews: see over
Dear editors:

Thank you for your letter. And I had rewritten the article according to your letter under the help of language experts. Innovation of this research work is that we found a new mutation with essential hypertension; further molecular biology studies confirm this locus mutations affect the stability of the tRNA and shear of tRNA, thereby affecting the tRNA binding polypeptide with an amino group. This work is a continuation of our previous work. As you mentioned 3 articles, they are our previous published articles and I had mentioned them in the article with no copyright conflicts. Finally, it is really a very long time since I first submitted this article to your journal. During this time, I had did revised the article according to two reviewer’s advice. Because of the time requirements of the subject, I hope to get your reply of the final decision about the article before the end 2013. Thank you for your help sincerely.

Best regards

Yuqi Liu